Author's response to reviews

Title: Is Provider Type Associated with Cancer Screening and Prevention? Advanced Practice Registered Nurses, Physician Assistants, and Physicians.

Authors:

Deanna L Kepka (deanna.kepka@hci.utah.edu)
Alexandria Smith (lexiasmith@gmail.com)
Christopher Zeruto (ZerutoC@imsweb.com)
K. Robin Yabroff (yabroffr@mail.nih.gov)

Version: 4 Date: 21 March 2014

Author's response to reviews:

Dear Editor,

We are very pleased with your most recent letter to our team. As requested, we have made the suggested changes to the attached manuscript.

1. We have clarified within the methods section of our manuscript that the data from the National Health Interview study are publicly available. This change is highlighted in yellow.

2. We have also clarified the funding sources for each author in our acknowledgement section. These changes are highlighted in yellow.

We greatly appreciate your time and interest.

Thank you kindly,

Deanna Kepka, PhD, MPH